Author's response to reviews

Title: 'Reaching the Hard to Reach' - Lessons learned from the VCS (Voluntary and Community Sector)

Authors:

Sarah M Flanagan (s.m.flanagan@bham.ac.uk)
Beverley Hancock (b.hancock@bham.ac.uk)

Version: 7 Date: 5 March 2010

Author's response to reviews:

Dear Editors,

MS: 1549536349235806

'Reaching the Hard to Reach' - Lessons learned from the VCS (Voluntary and Community Sector) Sarah M Flanagan and Beverley Hancock

Thank you for your comments on the above submission.

I have amended the manuscript accordingly -

1. I have added a little more detail to the analysis.

2. I have shortened the results section to focus on the results of this study and included the strengths and weaknesses in the discussion as suggested.

I have track marked the changes I have made as per instruction.

Many thanks for your consideration.

Yours sincerely

Sarah Flanagan